## WSR 06-04-034 EMERGENCY RULES DEPARTMENT OF REVENUE

[Filed January 26, 2006, 10:32 a.m., effective January 26, 2006]

Effective Date of Rule: Immediately.

Purpose: WAC 458-20-186 (Rule 186) provides tax-reporting information to persons who sell, use, consume, handle, possess, or distribute cigarettes. The rule explains who is liable for the tax, how and when the cigarette tax imposed by chapter 82.24 RCW is to be paid, and the record-keeping requirements. It explains the application process for whole-sale and retail cigarette vendor licenses, and includes references to statutory fees, bonding requirements, and explains the conditions for and process of application for a reinstatement of a license following a revocation under the Administrative Procedure Act.

The department is revising Rule 186 on an emergency basis to incorporate provisions of chapter 180, Laws of 2005 (SB 6097). This legislation made a number of changes with respect to the tobacco products tax program in chapter 82.26 RCW, and amended chapter 82.24 RCW to provide that any person possessing both a cigarette license and a tobacco products license is subject to suspension and revocation of both licenses for violation of either chapter 82.24 or 82.26 RCW.

Citation of Existing Rules Affected by this Order: Amending WAC 458-20-186 Tax on cigarettes.

Statutory Authority for Adoption: RCW 82.32.300 and 82.01.060(2).

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: An emergency adoption is necessary because a permanent rule cannot be adopted at this time. This rule action will provide needed information to tax-payers and department staff about the licensing requirements and responsibilities of persons selling cigarettes in this state. The rule being adopted is the same as that adopted on an emergency basis on September 29, 2005 (WSR 05-20-038).

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 1, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 0, Amended 1, Repealed 0.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 1, Repealed 0.

Date Adopted: January 26, 2006.

Janis P. Bianchi, Manager Interpretations and Technical Advice Unit

[AMENDATORY SECTION (Amending WSR 05-02-035, filed 12/30/04)]

WAC 458-20-186 Tax on cigarettes. (1) Introduction. This rule addresses those taxes and licensing activities that apply exclusively to cigarettes as defined by RCW 82.24.010. See WAC 458-20-185 for tax liabilities and registration requirements associated with tobacco products other than cigarettes. The tax on cigarettes is in addition to all other taxes owed. For example, retailers and wholesalers are liable for business and occupation tax on their retailing or wholesaling activities, and must collect and remit sales tax on retail sales of cigarettes. Consumers pay the cigarette tax in addition to sales or use tax on purchases of cigarettes for consumption within this state. (Wholesalers not licensed in the state of Washington who are making sales of cigarettes to Indians in accordance with a cigarette tax contract authorized by RCW 43.06.455 must comply with the specific terms of their individual contracts. See also WAC 458-20-192 regarding sales in Indian country.)

- (2) **Organization of rule.** The information provided in this rule is divided into seven parts:
- (a) Part I explains the tax liabilities of persons who sell, use, consume, handle, possess, or distribute cigarettes in this state.
- (b) Part II explains the licensing requirements and responsibilities for persons making wholesale or retail sales of cigarettes in this state.
- (c) Part III explains the stamping requirements and how the cigarette tax rates are calculated.
- (d) Part IV describes the exemptions from the tax and the procedures that must be followed to qualify for exemption.
- (e) Part V explains the requirements and responsibilities for persons transporting eigarettes in Washington.
- (f) Part VI explains the requirements and responsibilities for persons engaged in making delivery sales of cigarettes into this state.
- (g) Part VII explains the enforcement and administration of the cigarette tax.

#### Part I - Tax on Cigarettes

- (101) **In general.** The Washington state cigarette tax is due and payable by the first person who sells, uses, consumes, handles, possesses, or distributes the cigarettes in this state.
- (a) **Possession.** For the purpose of this rule, a "possessor" of cigarettes is anyone who personally or through an agent, employee, or designee, has possession of cigarettes in this state.
- (b) **Payment.** Payment of the cigarette tax is made through the purchase of stamps from banks authorized by the department of revenue (department) to sell the stamps. Only licensed wholesalers may purchase or obtain cigarette stamps. Except as specifically provided in Part IV of this rule, it is unlawful for any person other than a licensed whole-

[1] Emergency

saler to possess unstamped cigarettes in this state. However, as explained in subsection (102)(b) of this rule, certain consumers may possess unstamped cigarettes for personal consumption if they pay the tax as provided in this rule.

- (c) **Imposition of tax.** Ordinarily, the tax obligation is imposed on and collected from the first possessor of unstamped cigarettes. However, failure of an exempt entity with an obligation to collect and remit the tax does not relieve a subsequent nonexempt possessor of unstamped cigarettes from liability for the tax.
- (d) **Promotions.** Cigarettes given away for advertising or any other purpose are taxed in the same manner as if they were sold, used, consumed, handled, possessed, or distributed in this state, but are not required to have the stamp affixed. Instead, the manufacturer of the cigarettes must pay the tax on a monthly return filed with the department. See subsection (702) of this rule.

#### (102) Possession of cigarettes in Washington state.

- (a) Every person who is (i) in possession of unstamped cigarettes in this state, and (ii) is not specifically exempt by law, is liable for payment of the cigarette tax as provided in chapter 82.24 RCW and this rule.
- (b) Consumers who buy unstamped cigarettes or who purchase cigarettes from sources other than licensed retailers in this state must pay the cigarette tax as provided in subsection (702) of this rule when they first bring the cigarettes into this state or first possess them in this state. This requirement includes, but is not limited to, delivery sales as described in Part VI of this rule.
- (c) Cigarettes purchased from Indian retailers. Special rules apply to cigarettes purchased from Indian retailers.
- (i) Indians purchasing cigarettes in Indian country are exempt from the state cigarette tax; however, these sales must comply with WAC 458-20-192. Other consumers may purchase cigarettes for their personal consumption from "qualified Indian retailers" without incurring liability for state cigarette tax. A "qualified Indian retailer" is one who is subject to the terms of a valid cigarette tax contract with the state pursuant to RCW 43.06.455.
- (ii) Consumers who purchase cigarettes from Indian retailers who are not subject to a cigarette tax contract with the state must comply with the reporting requirements and remit the cigarette tax as explained in subsection (702) of this rule. These consumers are also liable for the use tax on their purchases. See WAC 458-20-178.
- (iii) It is the duty of the consumer in each instance to ascertain his or her responsibilities with respect to such purchases.
- (d) Cigarettes purchased on military reservations. Active duty or retired military personnel, and their dependants, may purchase cigarettes for their own consumption on military reservations without paying the state tax (see Part IV). However, such persons are not permitted to give or resell those cigarettes to others.
- (e) Counterfeit cigarettes. It is unlawful for any person to manufacture, sell, or possess counterfeit cigarettes. A cigarette is counterfeit if (i) it or its packaging bears any logo or marking used by a manufacturer to identify its own cigarettes, and (ii) the cigarette was not manufactured by the

owner of that logo or trademark or by any authorized licensee of the manufacturer. RCW 82.24.570.

(f) Possession of unstamped and untaxed cigarettes, and possession of counterfeit cigarettes, are criminal offenses in this state. See Part VII.

#### Part II - Wholesale and Retail Cigarette Vendor Licensing Requirements and Responsibilities

- (201) **License required.** No person, other than a government instrumentality or an Indian retailer as set forth in Part IV of this rule, may engage in the retail or wholesale distribution of cigarettes in this state without a license. No person may engage in the business of sampling within this state unless that person has first obtained a sampler's license. Failure to obtain the required license prior to sampling or selling cigarettes at wholesale or retail is a criminal act. RCW 70.155.050.
- (202) **Definitions.** For the purposes of this rule, the following definitions apply:
- (a) "Place of business" means any location where business is transacted with, or sales are made to, customers. The term includes, but is not limited to, any vehicle, truck, vessel, or the like at which sales are made.
- (b) "Retailer" means every person, other than a wholesaler, who purchases, sells, offers for sale, or distributes cigarettes, regardless of quantity or amount, or the number of sales, and all persons operating under a retailer's registration certificate.
- (c) "Retail selling price" means the ordinary, customary, or usual price paid by the consumer for each package of cigarettes, less the tax levied by the state.
- (d) "Wholesaler" means every person who purchases, sells, or distributes cigarettes, as defined in chapter 82.24 RCW, to retailers for the purpose of resale only.
- (203) **Wholesale license.** Prior to the sale or distribution of cigarettes at wholesale, each wholesaler must first obtain a wholesale cigarette license from the department of licensing.
- (a) **Background check.** Each wholesaler must undergo a criminal background check before a license will be issued. RCW 82.24.510. The background check must be completed to the satisfaction of the liquor control board and the department. Failure to provide information sufficient to complete the background check may, in the department's discretion, result in denial of the license.
- (b) **Application.** Application for license or renewal of license is made on forms supplied by the department of licensing and must be accompanied by the annual license fee as provided in chapter 82.24 RCW. A wholesale cigarette license is valid for one year from the date it is issued.
- (c) **Multiple locations.** If the wholesaler sells, or intends to sell, cigarettes at more than one place of business, whether temporary or permanent, a separate license with a license fee as provided in chapter 82.24 RCW is required for each additional place of business. Each license must be exhibited in the place of business for which it is issued.
- (d) **Bond required.** Each licensed wholesaler must file a bond with the department in an amount determined by the department, but not less than \$5,000.00. The bond must be executed by the wholesaler as principal, and by a corporation approved by the department of licensing and authorized to

Emergency [2]

engage in business as a surety company in this state, as surety. The bond must run concurrently with the wholesaler's license

### (204) Duties and responsibilities of licensed wholesalers.

- (a) **Stamps.** Only licensed wholesalers may purchase or obtain cigarette stamps. Wholesalers are prohibited by law from selling or providing stamps to any other wholesaler or person.
- (b) **Numbering.** Each roll of stamps, or group of sheets, has a separate serial number. The department keeps records of which wholesaler purchases each roll or group of sheets. Wholesalers are prohibited from possessing stamps other than those specifically issued to them.
- (c) **Sales restricted.** Wholesalers selling cigarettes in this state may sell cigarettes only to Washington retailers who have a current retail cigarette license, to other licensed wholesalers, or to Indian tribal entities authorized to possess cigarettes that are not taxed by the state.
- (d) **Unstamped cigarettes.** Except as explained in Part IV of this rule, no person other than a licensed wholesaler may possess unstamped cigarettes in this state. (For the purpose of this rule, the term "unstamped cigarette" means any cigarette that does not bear a Washington state cigarette stamp as described in Part III of this rule.) Licensed wholesalers may possess unstamped cigarettes in this state only in the following circumstances:
- (i) Licensed wholesalers may possess unstamped cigarettes for up to 72 hours after receipt; however, the cigarettes must be stamped on or before sale or transfer to any other party other than another licensed wholesaler. Licensed wholesalers may possess unstamped cigarettes for more than 72 hours after receipt if they receive prior written permission from the department to do so.
- (ii) Licensed wholesalers who have furnished a surety bond in an amount determined by the department may set aside, without stamping, that portion of their stock reasonably necessary for conducting sales to persons outside this state or to instrumentalities of the federal government. All unstamped stock must be kept separate and apart from stamped stock.
- (e) **Transfers.** Wholesalers in possession of unstamped cigarettes under subsection (204)(d) of this rule that are transferred by the wholesaler to another facility within this state must be transferred in compliance with RCW 82.24.250.
- (205) **Retail license.** Prior to the retail sale or distribution of cigarettes, each retailer must first be issued a retail cigarette license from the department of licensing. A license is required for each location at which cigarettes are sold at retail. Each license must be exhibited at the place of business for which it is issued.
- (a) **Application.** Applications for license or renewal of license are made on forms supplied by the department of licensing and must be accompanied by the annual license fee as provided in chapter 82.24 RCW. A retail cigarette license is valid for one year from the date it is issued.
- (b) **Vending machines.** Retailers operating cigarette vending machines are required to pay an additional annual fee as set forth in chapter 82.24 RCW for each vending machine.

#### (206) Duties and responsibilities of retailers.

- (a) No retailer in this state may possess unstamped cigarettes unless he or she is also a licensed wholesaler.
- (b) Retailers may obtain cigarettes only from cigarette wholesalers licensed by this state.
- (207) Additional requirements for manufacturers, wholesalers, retailers, and samplers. Persons making wholesale or retail sales or engaged in the business of sampling cigarettes in this state must comply with all the provisions of chapters 70.155 and 70.158 RCW. All cigarettes sold, delivered, or attempted to be delivered, in violation of RCW 70.155.105 are subject to seizure and forfeiture. RCW 82.24.130.

### (208) Suspension or revocation of wholesale or retail cigarette licenses.

- (a) The department has full power and authority to revoke or suspend the license of any wholesale or retail cigarette dealer in the state upon sufficient showing that the license holder has violated the provisions of chapter 82.24 RCW or this rule. See RCW 82.24.550 and WAC 458-20-10001 for information on the procedures pertaining to suspension or revocation of cigarette licenses.
- (b) Any person possessing both a cigarette license and a tobacco products license is subject to suspension and revocation of both licenses for violation of either chapter 82.24 or 82.26 RCW. For example, if a person has both a cigarette license and a tobacco license, revocation of the cigarette license will also result in revocation of the tobacco license.
- (c) A person whose license has been suspended or revoked must not sell or permit the sale of cigarettes or tobacco products on premises occupied or controlled by that person during the the period of the suspension or revocation.
- (d) For the purposes of this rule, "tobacco products" has the same meaning as in RCW 82.26.010.
- (<u>bd</u>) Any person whose license has been revoked must wait one year following the date of revocation before requesting a hearing for reinstatement. Reinstatement hearings are held pursuant to WAC 458-20-10001.

#### Part III - Stamping and Rates

#### (301) Cigarette stamps.

- (a) Stamps indicating payment of the cigarette tax must be affixed prior to any sale, use, consumption, handling, possession, or distribution of all cigarettes other than those specifically exempted as explained in Part IV of this rule. The stamp must be applied to the smallest container or package, unless the department, in its sole discretion, determines that it is impractical to do so. Stamps must be of the type authorized by the department and affixed in such a manner that they cannot be removed from the package or container without being mutilated or destroyed.
- (b) Licensed wholesalers may purchase state-approved cigarette stamps from authorized banks. Payment for stamps must be made at the time of purchase unless the wholesaler has prior approval of the department to defer payment and furnishes a surety bond equal to the proposed monthly credit limit. Payments under a deferred plan are due within 30 days following purchase. Licensed wholesalers are allowed a discount of \$6.00 per thousand stamps affixed ("stamping

[3] Emergency

allowance"), which amount is offset against the purchase price.

#### (302) Rates.

- (a) The Washington state cigarette tax is imposed on a per cigarette basis. The rate of the tax is a combination of statutory rates found in RCW 82.24.020, 82.24.027, and 82.24.028.
- (b) When the rate of tax increases, the first person who sells, uses, consumes, handles, possesses, or distributes previously taxed cigarettes after the rate increase is liable for the additional tax.
- (303) **Refunds.** Any person may request a refund of the face value of the stamps when the tax is not applicable and the stamps are returned to the department. Documentation supporting the claim must be provided at the time the claim for refund is made.
- (a) Refunds for stamped untaxed cigarettes sold to Indian tribal members or tribal entities will include the stamping allowance and will be approved by an agent of the department.
- (b) Refunds for stamped cigarettes will not include the stamping allowance if the stamps are:
- (i) Damaged, or unfit for sale, and as a result are destroyed or returned to the manufacturer or distributor; or
- (ii) Improperly or partially affixed through burns, jams, double stamps, stamped on carton flaps, or improperly removed from the stamp roll.
- (c) The claim for refund must be filed on a form provided by the department. An affidavit or a certificate from the manufacturer for stamped cigarettes returned to the manufacturer for destruction or by an agent of the department verifying the voiding of stamps and authorizing the refund must accompany the claim for refund.

#### Part IV - Exemptions

- (401) **In general.** There are limited exemptions from the cigarette tax provided by law. This part discusses exemptions and the procedures that must be followed to qualify for an exemption.
- (402) **Government sales.** The cigarette tax does not apply to the sale of cigarettes to:
- (a) The United States Army, Navy, Air Force, Marine Corps, or Coast Guard exchanges and commissaries and Navy or Coast Guard ships' stores;
  - (b) The United States Veteran's Administration; or
- (c) Any person authorized to purchase from the federal instrumentalities named in (a) or (b) above, if the cigarettes are purchased from the instrumentality for personal consumption.

#### (403) Sales in Indian country.

- (a) The definitions of "Indian," "Indian country," and "Indian tribe," in WAC 458-20-192 apply to this rule. "Cigarette contract" means an agreement under RCW 43.06.450 through 43.06.460.
- (b) The cigarette tax does not apply to cigarettes taxed by an Indian tribe in accordance with a cigarette contract under RCW 43.06.450 through 43.06.460.
- (c) The cigarette tax does not apply to cigarettes sold to an Indian in Indian country for personal consumption; however, those sales must comply with the allocation provisions

- of WAC 458-20-192. Sales made by an Indian cigarette outlet to nontribal members are subject to the tax, except as provided in (b) above.
- (d) See WAC 458-20-192 for information on making wholesale sales of cigarettes to Indians and Indian tribes.
- (404) Interstate commerce. The cigarette tax does not apply to cigarettes sold to persons licensed as cigarette distributors in other states when, as a condition of the sale, the seller either delivers the cigarettes to the buyer at a point outside this state, or delivers the same to a common carrier with the shipment consigned by the seller to the buyer at a location outside this state. Any person engaged in making sales to licensed distributors in other states or making export sales or in making sales to the federal government must furnish a surety bond in a sum equal to twice the amount of tax that would be affixed to the cigarettes that are set aside for the conduct of such business without affixing cigarette stamps. The unstamped stock must be kept separate and apart from any stamped stock.

#### Part V - Transporting Cigarettes in Washington

- (501) **Transportation of cigarettes restricted.** No person other than a licensed wholesaler may transport unstamped cigarettes in this state except as specifically set forth in RCW 82.24.250 and this rule, or as may be allowed under a cigarette tax contract subject to the provisions of RCW 43.06.455. Licensed wholesalers transporting unstamped cigarettes in this state must do so only in their own vehicles unless they have given prior notice to the liquor control board of their intent to transport unstamped cigarettes in a vehicle belonging to another person.
- (502) **Notice required.** Persons other than licensed wholesalers intending to transport unstamped cigarettes in this state must first give notice to the liquor control board of their intent to do so.
- (503) **Transportation of unstamped cigarettes.** All persons transporting unstamped cigarettes must have in their actual possession invoices or delivery tickets for such cigarettes. The invoices or delivery tickets must show the true name and address of the consignor or seller, the true name and address of the consignee or purchaser, and the quantity and brands of the cigarettes transported. It is the duty of the person responsible for the delivery or transport of the cigarettes to ensure that all drivers, agents, or employees have the delivery tickets or invoices in their actual possession for all such shipments.
- (504) **Consignment.** If the cigarettes transported pursuant to subsection (501), (502), or (503) of this rule are consigned to or purchased by any person in this state, that purchaser or consignee must be a person who is authorized by chapter 82.24 RCW to possess unstamped cigarettes in this state.
- (505) **Out-of-state shipments.** Licensed wholesalers shipping cigarettes to a point outside Washington or to a federal instrumentality must, at the time of shipping or delivery, report the transaction to the department. The report must show both (a) complete details of the sale or delivery, and (b) whether stamps have been affixed to the cigarettes.

The report may be made either by submitting a duplicate invoice or by completing a form provided by the department,

Emergency [4]

and must be filed with the department as set forth in subsection (702) of this rule.

(506) **Compliance required.** No person may possess or transport cigarettes in this state unless the cigarettes have been properly stamped or that person has fully complied with the requirements of RCW 82.24.250 and this rule. Failure to comply with the requirements of RCW 82.24.250 is a criminal act. Cigarettes in the possession of persons who have failed to comply are deemed contraband and are subject to seizure and forfeiture under RCW 82.24.130.

#### Part VI - Delivery Sales of Cigarettes

- (601) **Definitions.** The definitions in this subsection apply throughout this rule.
- (a) "Delivery sale" means any sale of cigarettes to a consumer in the state where either: (i) The purchaser submits an order for a sale by means of a telephonic or other method of voice transmission, mail delivery, any other delivery service, or the internet or other online service; or (ii) the cigarettes are delivered by use of mail delivery or any other delivery service. A sale of cigarettes made in this manner is a delivery sale regardless of whether the seller is located within or outside the state. (For example, "Royal Tax-free Smokes," located in the state of Vermont, offers sales via the internet and a toll-free telephone number, and ships its products to consumers in this state. These transactions are delivery sales.) A sale of cigarettes not for personal consumption to a person who is a wholesaler licensed under chapter 82.24 RCW or a retailer licensed under chapter 82.24 RCW is not a delivery sale.
- (b) "Delivery service" means any private carrier engaged in the commercial delivery of letters, packages, or other containers, that requires the recipient of that letter, package, or container to sign to accept delivery.
- (602) **Tax liability.** Cigarettes delivered in this state pursuant to a delivery sale are subject to tax as provided in Part I of this rule. Persons making delivery sales in this state are required to provide prospective consumers with notice that the sales are subject to tax pursuant to chapters 82.24 and 82.12 RCW, with an explanation of how the tax has been or is to be paid with respect to such sales.
- (603) **Additional requirements.** Persons making delivery sales of cigarettes in this state must comply with all the provisions of chapter 70.155 RCW. All cigarettes sold, delivered, or attempted to be delivered, in violation of RCW 70.155.105 are subject to seizure and forfeiture. RCW 82.24.-130.

#### Part VII - Enforcement and Administration

- (701) **Books and records.** An accurate set of records showing all transactions related to the purchase, sale, or distribution of cigarettes must be retained. RCW 82.24.090. These records may be combined with those required in connection with the tobacco products tax (see WAC 458-20-185), if there is a segregation therein of the amounts involved. All records must be preserved for five years from the date of the transaction.
- (702) **Reports and returns.** The department may require any person dealing with cigarettes in this state to complete and return forms, as furnished by the department,

- setting forth sales, inventory, and other data required by the department to maintain control over trade in cigarettes.
- (a) Manufacturers and wholesalers selling stamped, unstamped, or untaxed cigarettes must submit a complete record of sales of cigarettes in this state monthly. This report is due no later than the 15th day of the calendar month and must include all transactions occurring in the previous month.
- (b) Persons making sales of tax-exempt cigarettes to Indian tribes or Indian retailers pursuant to WAC 458-20-192 (9)(a) must transmit a copy of the invoice for each such sale to the special programs division of the department prior to shipment.
- (c) Wholesalers selling stamped cigarettes manufactured by nonparticipating manufacturers as defined in WAC 458-20-264 must report all such sales to the special programs division no later than the 25th day of the calendar month and must include all transactions occurring in the previous month.
- (d) Persons making sales of cigarettes into this state to other than a licensed wholesaler or retailer must file a report as required under Title 15, Chapter 10A, section 376 of the U.S. Code (commonly referred to as the "Jenkins Act" report). This report is due no later than the 10th day of each calendar month and must include all transactions occurring in the previous month.
- (e) Persons shipping or delivering any cigarettes to a point outside of this state must submit a report showing full and complete details of the interstate sale or delivery as set forth in Part V of this rule. This report is due no later than the 15th day of the calendar month immediately following the shipment or delivery.
- (f) Persons giving away unstamped cigarettes for advertising, promotional, or any other purpose, must report and pay the tax on the number of cigarettes distributed in this state
- (g) Consumers who buy unstamped cigarettes or who purchase cigarettes from sources other than licensed retailers in this state must pay the tax when they first bring the cigarettes into this state or first possess them in this state. The tax is paid with a "Tax Declaration for Cigarettes," which may be obtained from the department.
- (703) **Criminal provisions.** Chapter 82.24 RCW prohibits certain activities with respect to cigarettes. Persons handling cigarettes within this state must refer to these statutes. The prohibited activities include, but are not limited to, the following:
- (a) Transportation or possession of 60,000 or fewer cigarettes. Transportation or possession of 60,000 or fewer unstamped cigarettes is prohibited unless the notice requirements set forth in RCW 82.24.250 have been met; failure to meet those notice requirements is a gross misdemeanor. RCW 82.24.110 (1)(m).
- (b) **Transportation or possession of more than 60,000 cigarettes.** Transportation or possession of more than 60,000 unstamped cigarettes is prohibited unless the notice requirements set forth in RCW 82.24.250 have been met; failure to meet those notice requirements is a felony. RCW 82.24.110 (2).
- (c) **Forgery or counterfeiting of stamps.** Alteration, fabrication, forgery, and counterfeiting of stamps are felonies. RCW 82.24.100.

[5] Emergency

(d) **Counterfeit cigarettes.** The manufacture, sale, or possession of counterfeit cigarettes in this state is a felony. RCW 82.24.570.

(704) **Search, seizure, and forfeiture.** The department or the liquor control board may search for, seize, and subsequently dispose of unstamped cigarette packages and containers, counterfeit cigarettes, conveyances of all kinds (including aircraft, vehicles, and vessels) used for the transportation of unstamped and/or counterfeit cigarettes, and vending machines used for the sale of unstamped and/or counterfeit cigarettes. See RCW 82.24.130, et seq., for provisions relating to search, seizure, and forfeiture of property, possible redemption of property, and for treatment of such property in the absence of redemption.

(705) **Penalties.** RCW 82.24.120 provides a penalty for failure to affix the cigarette stamps or to cause the stamps to be affixed as required, or to pay any tax due under chapter 82.24 RCW. In addition to the tax deemed due, a penalty equal to the greater of \$10.00 per package of unstamped cigarettes or \$250.00 will be assessed. Interest is also assessed on the amount of the tax at the rate as computed under RCW 82.32.050(2) from the date the tax became due until the date of payment. The department may, in its sole discretion, cancel all or part of the penalty for good cause.

**Reviser's note:** The bracketed material preceding the section above was supplied by the code reviser's office.

**Reviser's note:** The typographical errors in the above section occurred in the copy filed by the agency and appear in the Register pursuant to the requirements of RCW 34.08.040.

#### WSR 06-04-036 EMERGENCY RULES DEPARTMENT OF REVENUE

[Filed January 26, 2006, 10:35 a.m., effective January 26, 2006]

Effective Date of Rule: Immediately.

Purpose: WAC 458-20-185 (Rule 185) explains the provisions of chapter 82.26 RCW, Tax on tobacco products. The rule identifies distributors who must pay the tax, the imposition of the tax, applicable penalties, the books and records that must be kept, and when a credit for previously paid tax may be taken.

The department is revising Rule 185 on an emergency basis to incorporate provisions of chapter 180, Laws of 2005 (SB 6097). This legislation made significant statutory changes to chapter 82.26 RCW, including:

- A requirement that distributors and retailers of tobacco products be licensed;
- Requiring that licensed distributors sell only to licensed retailers;
- Changing the measure of the tax from "wholesale value" to "taxable sales value";
- Imposing record-keeping requirements on persons that transport tobacco products; and
- Adding new enforcement provisions and transferring enforcement to the liquor control board.

Citation of Existing Rules Affected by this Order: Amending WAC 458-20-185 Tax on tobacco products.

Statutory Authority for Adoption: RCW 82.32.300 and 82.01.060(2).

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: An emergency adoption is necessary because a permanent rule cannot be adopted at this time. This rule action will provide needed information to tax-payers and department staff about the new licensing requirements and responsibilities of persons selling tobacco products in this state. The rule being adopted is the same as that adopted on an emergency basis on September 29, 2005 (WSR 05-20-037).

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 1, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 0, Amended 1, Repealed 0.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 1, Repealed 0.

Date Adopted: January 26, 2006.

Janis P. Bianchi, Manager Interpretations and Technical Advice Unit

[AMENDATORY SECTION (Amending WSR 03-12-058, filed 6/2/03)]

WAC 458-20-185 Tax on tobacco products. (1) Introduction. This rule explains the tax liabilities of persons engaged in business as a retailers, and/or distributors or subjobber of tobacco products. The tax on tobacco products is in addition to all other taxes owed. For example, retailers, and distributors, and subjobbers are liable for business and occupation tax on their retailing or wholesaling activities, use tax on tobacco products distributed as samples, and litter tax on the value of the tobacco products. See WAC 458-20-186 for tax liabilities associated with taxes which apply exclusively to cigarettes.

- (2) **Definitions.** The following definitions apply to this rule <u>unless the context clearly requires otherwise</u>.
- (a) "Tobacco products" means all tobacco products except cigarettes as defined in RCW 82.24.010. The term includes:
  - (i) Cigars, cheroots, stogies, and periques;
- (ii) Granulated, plug cut, crimp cut, ready rubbed, and other smoking tobacco;

Emergency [6]

- (iii) Snuff, snuff flour, cavendish, plug and twist tobacco, fine-cut, and other chewing tobaccos; and
- (iv) Shorts, refuse scraps, clippings, cuttings and sweepings of tobacco, and other kinds and forms of tobacco, prepared in such manner as to be suitable for chewing or smoking in a pipe or otherwise, or both for chewing and smoking.
- (b) "Manufacturer" means a person who manufactures and sells tobacco products.
  - (c) "Distributor" means:
- (i) Any person engaged in the business of selling tobacco products in this state who brings, or causes to be brought, into this state from without outside the state any tobacco products for sale:
- (ii) Any person who makes, manufactures, or fabricates, or stores tobacco products in this state for sale in this state;
- (iii) Any person engaged in the business of selling tobacco products without outside this state who ships or transports tobacco products to retailers in this state, to be sold by those retailers; or
- (iv) Any person engaged in the business of selling tobacco products in this state who handles for sale any tobacco products that are within this state but upon which tax has not been imposed.
- (d) "Subjobber" means any person, other than a tobacco manufacturer or distributor, who buys tobacco products from a distributor and sells them to persons other than the ultimate consumers.
- (ed) "Retailer" means any person engaged in the business of selling tobacco products to ultimate consumers.
- (fe) "Sale" means (iany) any transfer, exchange, or barter, in any manner or by any means whatsoever, for a consideration, and includes and means all sales made by any person. It (ii) The term "sale" includes all a gifts by a persons engaged in the business of selling tobacco products, for advertising, promoting, or as a means of evading the provisions of chapter 82.26 RCW., or for any other purposes whatsoever.
- (g) "Wholesale sales price" means the established price for which a manufacturer sells tobacco product to the distributor, exclusive of any discount or other reduction.
- (i) A wholesale sales price that is an established price must reflect the fair market value of the tobacco products. In the case where a seller and buyer establish a sales price that does not reflect fair market value, such as may occur in certain sales between affiliated companies, the wholesale sales price is the fair market value of the tobacco product and not the sales price established by the seller and buyer.
- (ii) The phrase "discount or other reduction" includes any reduction from the established wholesale sales price made to a specific customer or class of customers.

Example. Pursuant to a half-price promotion, a manufacturer sells tobacco products to a distributor. The invoice lists \$100 as the price of the product less a \$50 discount resulting in a net invoice of \$50. The tax is due on \$100 which is the wholesale sales price exclusive of any discount or other reduction.

 $(\frac{hf}{})$  "Business" means any trade, occupation, activity, or enterprise engaged in for the purpose of selling or distributing tobacco products in this state.

- (ig) "Place of business" means any place where tobacco products are sold or where tobacco products are manufactured, stored, or kept for the purpose of sale or consumption, including any vessel, vehicle, airplane, train, or vending machine.
- (jh) "Retail outlet" means each place of business from which tobacco products are sold to consumers.
  - $(\underline{ki})$  "Department" means the department of revenue.
- (‡j) "Person" means any individual, receiver, administrator, executor, assignee, trustee in bankruptcy, trust, estate, firm, copartnership, joint venture, club, company, joint stock company, business trust, municipal corporation, the state and its departments and institutions, political subdivision of the state of Washington, corporation, limited liability company, association, society, or any group of individuals acting as a unit, whether mutual, cooperative, fraternal, nonprofit, or otherwise. The term excludes any person immune from state taxation, including the United States or its instrumentalities, and federally recognized Indian tribes and enrolled tribal members, conducting business within Indian country.
- (mk) "Indian country" means the same as defined in WAC 458-20-192 chapter 82.24 RCW.
- (1) "Actual price" means the total amount of consideration for which tobacco products are sold, valued in money, whether received in money or otherwise, including any charges by the seller necessary to complete the sale such as charges for delivery, freight, transportation, or handling.
- (m) "Affiliated" means related in any way by virtue of any form or amount of common ownership, control, operation, or management.
  - (n) "Board" means the liquor control board.
- (o) "Cigar" means a roll for smoking that is of any size or shape and that is made wholly or in part of tobacco, irrespective of whether the tobacco is pure or flavored, adulterated or mixed with any other ingredient, if the roll has a wrapper made wholly or in greater part of tobacco. "Cigar" does not include a cigarette.
- (p) "Cigarette" has the same meaning as in RCW 82.24.010.
- (q) "Manufacturer's representative" means a person hired by a manufacturer to sell or distribute the manufacturer's tobacco products, and includes employees and independent contractors.
  - (r) "Taxable sales price" means:
- (i) In the case of a taxpayer that is not affiliated with the manufacturer, distributor, or other person from whom the taxpayer purchased tobacco products, the actual price for which the taxpayer purchased the tobacco products;
- (ii) In the case of a taxpayer that purchases tobacco products from an affiliated manufacturer, affiliated distributor, or other affiliated person, and that sells those tobacco products to unaffiliated distributors, unaffiliated retailers, or ultimate consumers, the actual price for which that taxpayer sells those tobacco products to unaffiliated distributors, unaffiliated retailers, or ultimate consumers;
- (iii) In the case of a taxpayer that sells tobacco products only to affiliated distributors or affiliated retailers, the price, determined as nearly as possible according to the actual price, that other distributors sell similar tobacco products of like

[7] Emergency

quality and character to unaffiliated distributors, unaffiliated retailers, or ultimate consumers;

- (iv) In the case of a taxpayer that is a manufacturer selling tobacco products directly to ultimate consumers, the actual price for which the taxpayer sells those tobacco products to ultimate consumers;
- (v) In the case of a taxpayer that has acquired tobacco products under a sale as defined in subsection (2)(e)(ii) of this rule, the price, determined as nearly as possible according to the actual price, that the taxpayer or other distributors sell the same tobacco products or similar tobacco products of like quality and character to unaffiliated distributors, unaffiliated retailers, or ultimate consumers; or
- (vi) In any case where (i) through (v) of this subsection do not apply, the price, determined as nearly as possible according to the actual price, that the taxpayer or other distributors sell the same tobacco products or similar tobacco products of like quality and character to unaffiliated distributors, unaffiliated retailers, or ultimate consumers.

For purposes of (i) and (ii) of this subsection only, "person" includes both persons as defined in subsection (2)(j) of this rule and any person immune from state taxation, including the United States or its instrumentalities, and federally recognized Indian tribes and enrolled tribal members, conducting business within Indian country.

- (s) "Taxpayer" means a person liable for the tax imposed by chapter 82.26 RCW.
- (t) "Unaffiliated distributor" means a distributor that is not affiliated with the manufacturer, distributor, or other person from whom the distributor has purchased tobacco products.
- (u) "Unaffiliated retailer" means a retailer that is not affiliated with the manufacturer, distributor, or other person from whom the retailer has purchased tobacco products.
- (3) Distributor and retail tobacco products vendor licensing requirements and responsibilities.
- (a) License required. No person, other than a government instrumentality or an Indian retailer as set forth in subsection of this rule, may engage in the retail or wholesale distribution of tobacco products in this state without a license.
- (b) Distributor's license. Prior to selling or distributing tobacco products from a stock of goods in Washington or selling to retailers in Washington, each distributor must first obtain a tobacco distributor's license from the department of licensing.
- (i) Background check. Each distributor must undergo a criminal background check before a license will be issued. Chapter 82.26 RCW. The background check must be completed to the satisfaction of the liquor control board and the department of revenue. Failure to provide information sufficient to complete the background check may result in denial of the license. A background check will not be required if the applicant has had a background check under chapter 66.24 or 82.24 RCW.
- (ii) Application. Application for a license or renewal of license is made on forms supplied by the department of licensing and must be accompanied by the annual license fee as provided in chapter 82.26 RCW. A distributor license is valid for one year from the date it is issued. The annual fees

- will not apply if the licensee pays the corresponding annual cigarette distributor fees under RCW 82.24.510.
- (iii) Multiple locations. If the distributor sells, or intends to sell, tobacco products at more than one place of business, whether temporary or permanent, a separate license with a license fee as provided in chapter 82.26 RCW is required for each additional place of business. Each license must be exhibited in the place of business for which it is issued.

### (c) Duties and responsibilities of licensed distributors.

- (i) Sales restricted. Distributors selling tobacco products in this state may sell those products only to Washington retailers or distributors who have a current tobacco products license, to other licensed distributors, or to Indian tribal entities authorized to possess tobacco products that are not taxed by the state.
- (ii) Manufacturer's representatives. Manufacturers selling tobacco products through manufacturer's representatives must provide the department a current list of the names, addresses, and telephone numbers of all such representatives. Mail the list to Washington State Department of Revenue, PO Box 47477, Olympia, WA 98504. The manufacturer must have a distributor's license and its representatives must carry a copy of the license at all times when selling or distributing tobacco products in this state.
- (d) Retail license. Prior to the retail sale or distribution of tobacco products, each retailer must first be issued a retail tobacco license from the department of licensing. A license is required for each location at which tobacco products are sold at retail. Each license must be exhibited at the place of business for which it is issued.

Applications for a license or the renewal of a license are made on forms supplied by the department of licensing and must be accompanied by the annual license fee as provided in chapter 82.26 RCW. A retail tobacco license is valid for one year from the date it is issued. The annual tobacco license fees will not apply if the licensee pays the corresponding annual cigarette retailer license fees as provided in RCW 82.24.510.

- (e) Duties and responsibilities of retailers. A retailer that obtains tobacco products from an unlicensed distributor or any other person that is not licensed under chapter 82.26 RCW must be licensed both as a retailer and a distributor. The retailer is liable for the tax imposed under RCW 82.26.020 with respect to the tobacco products acquired from the unlicensed person that are held for sell, handling, or distribution in this state. For example, if a retailer buys tobacco products from an Indian smoke shop or an out of state distributor that does not have a tobacco distributor license, the retailer must obtain a distributor license and pay the tobacco tax due.
- (34) Rate and measure of tax. The Washington state tobacco products tax is an excise tax levied on the wholesale taxable sales price (as defined in RCW 82.26.010 and this rule) on all tobacco products sold, used, consumed, handled, or distributed within the state.

The rate of tax is a combination of statutory percentage rates found in RCW 82.26.020, 82.26.025, and 82.26.028. The total current rate of tax is shown on the current combined excise tax return.

Emergency [8]

- (45) **Imposition of tax.** The tax is imposed once on all tobacco products sold, used, consumed, handled, or distributed within this state.
- (a) **When tax is imposed.** The tax is imposed at the time the distributor:
- (i) Brings, or causes to be brought, into this state from without outside the state tobacco products for sale; or
- (ii) Makes, manufacturers, or fabricates tobacco products in this state for sale in this state; or
- (iii) Ships or transports tobacco products to retailers in this state, to be sold by those retailers; or
- (iv) Handles for sale any tobacco products that are within this state but upon which tax has not been imposed. For example, a retailer with a place of business in this state purchases for sale tobacco products from an enrolled tribal member of a federally recognized tribe located within Indian country. Because the tax was not imposed on the enrolled tribal member, the retailer must also be licensed as a distributor pay and pay the tax.
- (b) Additional occasion when tax may be imposed. Any retailer who fails to keep invoices as required under chapter 82.32 RCW and which invoices do not conform to the requirements set forth in subsection (5)(b) of this rule is liable for the tax on any uninvoiced tobacco product which that retailer handles for sale.
- (c) When an out-of-state person is a distributor who must pay the tax. A person located out-of-state who is selling tobacco products to Washington wholesalers from a stock of goods located outside this state is not a distributor and therefore is not liable for the tax.
- (i) On the other hand, a person located out-of-state who is selling and shipping tobacco products to Washington retailers from an out-of-state stock of goods is a distributor and is subject to the tax. If the out-of-state person is not required to register and pay taxes in Washington, the retailers to whom it sells must pay the tax. However, such out-of-state persons may elect to register with the state and pay the tax.
- (ii) A Washington retailer who purchases tobacco from an out-of-state stock of goods from a person located out-of-state who is not required to register and pay taxes in Washington may provide to that person a certificate affirming that the Washington retailer will remit to the state the tax due. Both the out-of-state person and the Washington retailer should retain a copy of such certificate. The certificate should substantially conform to the example shown below:

#### Retailer's Certificate of Remittance of Tax

The undersigned retailer hereby certifies that the undersigned will remit to the state the tax due on the tobacco products specified below purchased from seller. This certificate shall be considered a part of each order which the undersigned may give to the seller, unless otherwise specified, and shall be valid until revoked by the undersigned in writing or until it expires, whichever occurs first. This certificate expires four years from the effective date.

Name of Seller:	
Name of Retailer	Effective Date
UBI/Registration #	
Address of Retailer	

Tobacco products purchased	-	•			•		 ٠	•	•	•	•	•	•		
Agent for Retailer (print)		•		 •	•			•	•					•	
Signature															

- (iii6) Tax on samples and sales to dual purchasers. (i) A person who is located out-of-state and who is required to register and pay taxes in Washington may sell and ship tobacco products to a Washington customer who is both a wholesaler distributor and retailer. Under this circumstance, the person seller, the customer buyer, and the department may enter into a written agreement that identifies the person who will remit to the state the tax due as to those particular sales. The written agreement will contain such other terms and conditions that are acceptable to the department.
- (<u>ivii</u>) When a person located outside Washington distributes samples in this state, that person must pay the tax on those samples.
- (5) Books and records. Since the tobacco products tax is paid on returns as computed by the taxpayer rather than by affixing of stamps or decals, the law contains stringent provisions requiring that accurate and complete records be maintained. The records must include all pertinent papers and documents relating to the purchase, sale, or disposition of tobacco products and must be kept for a period of at least five years after the date of the document or the date of the entry appearing in the records.

#### (7) Record keeping and enforcement.

- (a) Books and records. An accurate set of records showing all transactions related to the purchase, sale, or distribution of tobacco products must be retained for a period of at least five years. RCW 82.26.060, 82.26.070, and 82.26.080.
- (ab) **Distributors.** Distributors must keep at each registered place of business complete and accurate records for that place of business. The records to be kept by distributors include itemized invoices of tobacco products held, purchased, manufactured, brought in or caused to be brought in from without outside the state or shipped or transported to retailers in this state, and of all sales of tobacco products products except retail sales. The itemized invoice for each purchase or sale must be legible and must show the seller's name and address, the purchaser's name and address, the date of sale, and all prices and discounts. Itemized invoices must be preserved for five years from the date of sale.
- (bc) Retailers Retailers and subjobbers. Retailers and subjobbers must secure itemized invoices of all tobacco products purchased. The itemized invoice for each purchase must be legible and must show the seller's name and address, the purchaser's name and address, the date of sale, and all prices and discounts. Itemized invoices must be preserved for five years from the date of sale.
- (e) Warehouses. Records of all deliveries or shipments (including ownership, quantities) of tobacco products from any public warehouse of first destination in this state must be kept by the warehouse.
- (6) Nonpayment of tax by retailers. If the department finds that any nonpayment of tax by the retailer was willful, penalties and interest shall be assessed in accordance with chapter 82.32 RCW. In the case of a second or plural nonpayment of tax by the retailer, penalties and interest will be

[9] Emergency

assessed in accordance with chapter 82.32 RCW without regard to willfulness.

- (a) Example. In the course of an audit of Retailer, the department determines that on several occasions Retailer failed to pay the tax. The department does not find the non-payment to be willful. Retailer owes the tax due on all occasions of nonpayment and the penalties and interest is assessed on all but the first occasion of nonpayment. A few years later Retailer is audited again. The department finds one occasion of nonpayment of tax. In addition to the tax due, penalties and interest will be assessed in accordance with chapter 82.32 RCW.
- (b) Example. In the course of an audit of Retailer #2, the department determines that on several occasions Retailer #2 failed to pay the tax. The department determines that the non-payment of tax was willful. In addition to the tax due on all occasions of nonpayment, Retailer #2 owes penalties and interest on all occasions.
- (7<u>d</u>) **Reports and returns.** ((The tax is reported on the combined excise tax return to be filed according to the reporting frequency assigned by the department. Detailed instructions for preparation of these returns may be secured from the department.

Out of state wholesalers or distributors selling directly to retailers in Washington should apply for a certificate of registration, and the department will furnish returns for reporting the tax.

Retailers, distributors, and subjobbers may be required to file a report with the department in compliance with the provisions of the National Uniform Tobacco Settlement when purchasing tobacco products from certain manufacturers. Please see WAC 458-20-264 and chapter 70.157 RCW.)) (i) The department may require any person dealing in tobacco products in this state to complete and return forms, as furnished by the department, setting forth sales, inventory, and other data required by the department to maintain control over trade in tobacco.

- (ii) **Tax Returns.** Every distributor shall report the tax on an excise tax return as provided in RCW 82.32.045.
- (e) <u>Criminal provisions</u>. Chapter 82.26 RCW prohibits certain activities with respect to tobacco products. Persons handling tobacco within this state must refer to these statutes.
- (f) Search, seizure, and forfeiture. Any tobacco products in the possession of a person selling tobacco in this state without a license or transporting tobacco products without the proper invoices or delivery tickets may be seized without a warrant by any agent of the department, agent of the board, or law enforcement officer of this state. In addition, all conveyances, including aircraft, vehicles, or vessels, used to transport the illegal tobacco product may be seized and forfeited.
- (g) Enforcement. Pursuant to RCW 82.26.121 and 66 44.010, enforcement officers of the liquor control board may enforce all provisions of the law with respect to the tax on tobacco products. Retailers and distributors must allow department personnel and enforcement officers of the liquor control board free access to their premises to inspect the tobacco products contained in the premises and to examine the books and records of the business. Failure to allow free access or to hinder or interfere with department personnel

- and/or enforcement officers of the liquor control board may result in revocation of the tobacco products license.
- (h) Suspension or revocation of distributor or retail tobacco licenses. (i) The department has full power and authority to revoke or suspend the license of any wholesale or retail tobacco dealer in the state upon sufficient showing that the license holder has violated the provisions of chapter 82.26 RCW or this rule. If a person has both a tobacco license and a cigarette license, revocation of the tobacco license will also result in revocation of the cigarette license.
- (ii) Any person whose license has been revoked must wait one year following the date of revocation before requesting a hearing for reinstatement. Reinstatement hearings are held pursuant to WAC 458-20-10001.
- (i) **Penalties.** Penalties and interest may be assessed in accordance with chapter 82.32 RCW for nonpayment of tobacco tax.
  - (8) Transporting tobacco products in Washington.
- (a) Transportation of tobacco products restricted. No person other than a licensed distributor, an authorized manufacturer's representative, or a licensed retailer may transport tobacco products for sale in this state, except as specifically set forth in chapter 82.26 RCW. Licensed distributors and retailers transporting tobacco products in this state must do so only in their own vehicles unless they have given prior notice to the liquor control board of their intent to transport tobacco products in a vehicle belonging to another person.
- (b) Notice required. Persons other than licensed distributors, licensed retailers, or authorized manufacturer's representatives intending to transport tobacco products in this state must first give notice to the liquor control board of their intent to do so.
- (c) Invoices required. All persons transporting tobacco products for sale in this state must have in their actual possession invoices or delivery tickets for the tobacco products that show the true name and address of the consignor or seller, the true name and address of the consignee or buyer, and the quantity and brands of the tobacco products being transported. Persons having tobacco products transported in this state on their behalf by a third party must insure that the persons transporting the tobacco products have the required invoices or delivery tickets in their possession.
- (d) Inspection. In any case where the department or the board, or any law enforcement officer of the state, has knowledge or reasonable grounds to believe that any vehicle is transporting tobacco products in violation of chapter 82.26 RCW or this rule, the department, board, or law enforcement officer is authorized to stop the vehicle and inspect it for contraband tobacco products.

#### (89) Interstate sales and sales to U.S.

(a) The tax does not apply to tobacco products sold to federal government agencies, nor to deliveries to retailers outside the state for resale resale by such retailers, and a credit may be taken for the amount of tobacco products tax previously paid on such products. RCW 82.26.110. The credit is not available for sales made for delivery outside this state other than sales for resale resale to retailers. For example, no credit may be taken for a sale of tobacco products delivered to a consumer outside the state.

Emergency [10]

(b) To document that the tobacco products were sold to a retailer outside the state for resale by such retailer, the person seller may obtain from the retailer buyer a certificate which that substantially conforms to the following:

Retailer's Certification of Purchase of Tobacco Products for Resale Outside Washington

The undersigned buyer/retailer hereby certifies that the tobacco products specified below are purchased for resale outside this state by the undersigned. This certificate shall be considered a part of each order which the undersigned may give to the seller, unless otherwise specified, and shall be valid until revoked in writing by the undersigned or until it expires, whichever occurs first. This certificate expires four years from the effective date.

Name of Seller	Effective Date
UBI/Registration #	
Name of Buyer/Retailer Business	s
Address	
Items purchased for resale	
Agent for buyer/retailer (print)	
Signature	

(910) **Returned or destroyed goods.** A credit may also be taken for tobacco products destroyed or returned to the manufacturer on which tax was previously paid. If the credit is claimed against tax owed by the taxpayer or as a refund of tax paid, taxpayers must retain in their records appropriate documentation, affidavits or certificates conforming to those illustrated below:

#### (a) Certificate of taxpayer.

Claim for Credit on Tobacco Products
Tax Merchandise Destroyed

The undersigned certifies under penalty of perjury under the laws of the state of Washington that the following is true and correct to the best of his/her knowledge:

That he/she is \_\_(Title)\_ of the \_\_(Business Name)\_, a dealer in tobacco products; that the dealer has destroyed merchandise unfit for sale, said tobacco products having a wholesale sales price of \$ . . . . . ; that tobacco tax had been paid on such tobacco products; that the tobacco products were destroyed in the following manner and in the presence of an authorized agent of the department of revenue:

(State date	and manner of destruction)
Attested to: Date	By
	Position with Dealer
	Dealer

	Address of Dealer
APPROVED:	
Authorized Agent of	
Department of Revenue of the	
State of Washington.	

#### (b) Certificate of manufacturer.

Claim for Credit on Tobacco Products
Tax Merchandise Returned:

The undersigned certifies under penalty of perjury under the laws of the state of Washington that the following is true and correct to the best of his/her knowledge:

That he/she is \_\_(Title) of the \_\_(Business Name)\_\_, a manufacturer of tobacco products; that the manufacturer has received from \_\_(Dealer)\_\_, \_\_(Address)\_\_, a dealer in tobacco products within the State of Washington, certain tobacco products which were unfit for sale, the tobacco products having a wholesale sales price of \$ . . . . . .; that the tobacco products were destroyed in the following manner:

(Indicate date and m	nanner of destruction)
Credit issued on Memo No. credit approved by:	Signature of Taxpayer or Authorized Representative
on behalf of the Department of Revenue - State of	Name of Manufacturer
Washington	Address

(10) Enforcement. Pursuant to RCW 82.26.121 and 66.44.010, enforcement officers of the liquor control board may enforce all provisions of the law with respect to the tax on tobacco products. Retailers, distributors, and subjobbers must allow department personnel and enforcement officers of the liquor control board free access to their premises to inspect the tobacco products contained in the premises and to examine the books and records of the business. Failure to allow free access or to hinder or interfere with department personnel and/or enforcement officers of the liquor control board may result in the revocation of the business license.

**Reviser's note:** The bracketed material preceding the section above was supplied by the code reviser's office.

**Reviser's note:** The typographical errors in the above section occurred in the copy filed by the agency and appear in the Register pursuant to the requirements of RCW 34.08.040.

[11] Emergency

## WSR 06-05-014 EMERGENCY RULES DEPARTMENT OF FISH AND WILDLIFE

[Order 06-19—Filed February 3, 2006, 10:48 a.m., effective February 3, 2006]

Effective Date of Rule: Immediately. Purpose: Amend personal use rules.

Citation of Existing Rules Affected by this Order: Repealing WAC 232-28-61900G; and amending WAC 232-28-619.

Statutory Authority for Adoption: RCW 77.12.240.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: The upriver fishing boundary on the Okanogan River open to the harvest of all game fish was incorrectly identified as the "Highway 97 Bridge at Malott." The correct location is the "Highway Bridge at Malott." There is insufficient time to promulgate permanent rules.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 1, Amended 0, Repealed 1.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 0, Repealed 0.

Date Adopted: February 2, 2006.

J. P. Koenings Director by Larry Peck

#### **NEW SECTION**

WAC 232-28-619001 Exceptions to statewide rules—Columbia, Methow, Okanogan and Similkameen rivers. Notwithstanding the provisions of WAC 232-28-619, effective immediately until further notice, it is unlawful to violate the following provisions in the following waters:

- (1) For purposes of this section, "adipose fin clipped steelhead" means steelhead with an adipose fin clip and a healed scar at the site of the fin clip, whether or not any other fins are clipped or a healed scar is present at any other fin position.
- (2) Columbia River from Highway 395 Bridge at Pasco to the Old Hanford townsite wooden powerline towers upstream of Ringold Hatchery. Daily limit may contain up to two adipose fin clipped steelhead.

- (3) Columbia River from Rocky Reach Dam to 400 feet below Chief Joseph Dam - Open until further notice. Night closure in effect. Daily limit may contain up to two adipose fin clipped steelhead.
- (4) Okanogan River Open until further notice, except closed from Lake Osoyoos Control Dam (Zosel Dam) downstream to one-quarter mile below railroad trestle. Selective gear rules except lawful to fish from motorized vessels. Night closure in effect. Gamefish: Open to all gamefish downstream from Highway Bridge at Malott. Daily limit may contain up to two adipose fin clipped steelhead. Above Highway Bridge at Malott, open only for adipose fin clipped steelhead. Daily limit may contain up to two adipose fin clipped steelhead.
- Except: Effective March 1, 2006 until further notice, the area from the Highway 97 Bridge at Omak to a line across the river 500 feet above the mouth of Omak Creek, closed to fishing.
- (5) Similkameen River Mouth to 400 feet below Enloe Dam Open until further notice. Selective gear rules. Night closure in effect. All species: Release all fish except up to two adipose fin clipped steelhead per day may be retained and whitefish may be retained. Whitefish gear rules do not apply.
- (6) Methow River Open until further notice. Mouth (Highway 97 Bridge) upstream to the second powerline crossing, and from the first Highway 153 Bridge north of Pateros to the confluence with the Chewuch River. Selective gear rules except lawful to fish from motorized vessels. Night closure in effect. All species: Release all fish except up to two adipose fin clipped steelhead per day may be retained and whitefish may be retained. Whitefish gear rules do not apply.

#### **REPEALER**

The following section of the Washington Administrative Code is repealed:

WAC 232-28-61900G

Exceptions to statewide rules—Columbia, Methow, Okanogan and Similkameen rivers. (06-15)

## WSR 06-05-015 EMERGENCY RULES DEPARTMENT OF SOCIAL AND HEALTH SERVICES

(Health and Recovery Services Administration) [Filed February 3, 2006, 11:45 a.m., effective February 7, 2006]

Effective Date of Rule: February 7, 2006.

Purpose: WAC 388-538-063 is being amended to remove incorrect authorization language.

Citation of Existing Rules Affected by this Order: Amending WAC 388-538-063.

Statutory Authority for Adoption: RCW 74.08.090, 74.09.035, 74.09.522.

Emergency [12]

Other Authority: ESSB 5404, Section 209(15), chapter 25, Laws of 2003 1st sp.s.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: This emergency rule adoption is necessary while the permanent rule-making process is being completed. The current WAC incorrectly directs the client to receive authorization for services from the managed care organization (MCO) instead of the department. This incorrect language causes confusion and delays in the delivery of medically necessary services and results in the loss of due process rights for the enrollee. As the WAC now reads, enrollees must exhaust the MCO's appeal process before they can file a fair hearing. For GAU clients, the MCO does not have an appeal process because it is the department that approves or denies services. This creates a "catch twenty-two" for enrollees who have no redress through the MCO, but also cannot file a fair hearing request until the rule is corrected.

This continues the emergency rule that is currently in effect under WSR 05-21-027. The permanent rule has been adopted under WSR 06-03-081 and will supercede this emergency rule when the permanent rule becomes effective February 12, 2006. This emergency rule is necessary to cover the gap between the expiration of the current emergency rule (February 6) and the effective date of the permanent rule.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 1, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 0, Amended 0, Repealed 0.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 1, Repealed 0.

Date Adopted: January 25, 2006.

Andy Fernando, Manager Rules and Policies Assistance Unit

AMENDATORY SECTION (Amending WSR 04-15-003, filed 7/7/04, effective 8/7/04)

WAC 388-538-063 Mandatory enrollment in managed care for GAU clients. (1) The purpose of this section is to describe the managed care requirement for general assistance unemployable (GAU) clients mandated by the Laws of 2003, chapter 25, section 209(15).

- (2) The only sections of chapter 388-538 WAC that apply to GAU clients described in this section are incorporated by reference into this section.
- (3) To receive medical assistance administration (MAA) paid medical care, GAU clients must enroll in a managed care plan as required by WAC 388-505-0110(7) when they reside in a county designated as a mandatory managed care plan county.
- (4) GAU clients are exempt from mandatory enrollment in managed care if they:
  - (a) Are American Indian or Alaska Native (AI/AN); and
- (b) Meet the provisions of 25 U.S.C. 1603 (c)-(d) for federally recognized tribal members and their descendants.
- (5) In addition to subsection (4), MAA will exempt a GAU client from mandatory enrollment in managed care or end an enrollee's enrollment in managed care in accordance with WAC 388-538-130(3) and 388-538-130(4).
- (6) On a case-by-case basis, MAA may grant a GAU client's request for exemption from managed care or a GAU enrollee's request to end enrollment when, in MAA's judgment:
- (a) The client or enrollee has a documented and verifiable medical condition; and
- (b) Enrollment in managed care could cause an interruption of treatment that could jeopardize the client's or enrollee's life or health or ability to attain, maintain, or regain maximum function.
- (7) MAA enrolls GAU clients in managed care effective on the earliest possible date, given the requirements of the enrollment system. MAA does not enroll clients in managed care on a retroactive basis.
- (8) Managed care organizations (MCOs) that contract with MAA to provide services for GAU clients must meet the qualifications and requirements in WAC 388-538-067 and 388-538-095 (3)(a), (b), (c), and (d).
- (9) MAA pays MCOs capitated premiums for GAU enrollees based on legislative allocations for the GAU program.
- (10) GAU enrollees are eligible for the scope of care as described in WAC 388-529-0200 for medical care services (MCS). Other scope of care provisions that apply:
- (a) A client is entitled to timely access to medically necessary services as defined in WAC 388-500-0005;
- (b) MCOs cover the services included in the managed care contract for GAU enrollees. MCOs may, at their discretion, cover services not required under the MCO's contract for GAU enrollees:
- (c) MAA pays providers on a fee-for-service basis for the medically necessary, covered medical care services not covered under the MCO's contract for GAU enrollees; <u>and</u>
- (d) ((Even if a service is covered by MAA on a fee-forservice basis, it is the MCO, and not MAA, from whom a GAU enrollee must obtain prior authorization before receiving the service; and
- (e))) A GAU enrollee may obtain emergency services in accordance with WAC 388-538-100.
- (11) MAA does not pay providers on a fee-for-service basis for services covered under the MCO's contract for GAU enrollees, even if the MCO has not paid for the service,

[13] Emergency

regardless of the reason. The MCO is solely responsible for payment of MCO-contracted health care services that are:

- (a) Provided by an MCO-contracted provider; or
- (b) Authorized by the MCO and provided by nonparticipating providers.
- (12) The following services are not covered for GAU enrollees unless the MCO chooses to cover these services at no additional cost to MAA:
  - (a) Services that are not medically necessary;
- (b) Services not included in the medical care services scope of care;
- (c) Services, other than a screening exam as described in WAC 388-538-100(3), received in a hospital emergency department for nonemergency medical conditions; and
- (d) Services received from a nonparticipating provider requiring prior authorization from the MCO that were not authorized by the MCO.
- (13) A provider may bill a GAU enrollee for noncovered services described in subsection (12), if the requirements of WAC 388-502-0160 and 388-538-095(5) are met.
- (14) The grievance and appeal process found in WAC 388-538-110 applies to GAU enrollees described in this section.
- (15) The ((fair)) hearing process found in chapter 388-02 WAC and WAC 388-538-112 applies to GAU enrollees described in this section.

## WSR 06-05-021 EMERGENCY RULES DEPARTMENT OF SOCIAL AND HEALTH SERVICES

(Health and Recovery Services Administration) [Filed February 6, 2006, 4:21 p.m., effective February 7, 2006]

Effective Date of Rule: February 7, 2006.

Purpose: The department is amending these rules to reconcile SSI-related medical program resource eligibility rules with federal law.

Citation of Existing Rules Affected by this Order: Amending WAC 388-475-0550, 388-475-0700, 388-475-0800, 388-475-0820, and 388-475-0860.

Statutory Authority for Adoption: RCW 74.04.050, 74.04.057, 74.08.090, and 74.09.500.

Other Authority: Social Security Act as amended by P.L. 108-203.

Under RCW 34.05.350 the agency for good cause finds that state or federal law or federal rule or a federal deadline for state receipt of federal funds requires immediate adoption of a rule.

Reasons for this Finding: The state must adopt rules to comply with federal Medicaid law in Public Law 108-203, Subtitle D. Sections 430, 431, and 435 in order to continue receiving federal funding. This continues the emergency rule adopted under WSR 05-21-026 on October 10, 2005. The permanent rule has been adopted under WSR 06-04-046 and will supersede this emergency rule when the permanent rule becomes effective February 26, 2006. This emergency rule is necessary to cover the gap between the expiration of the cur-

rent emergency rule (midnight on February 7) and the effective date of the permanent rule.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 5, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 0, Amended 0, Repealed 0.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 5, Repealed 0.

Date Adopted: February 2, 2006.

Andy Fernando, Manager Rules and Policies Assistance Unit

AMENDATORY SECTION (Amending WSR 04-09-004, filed 4/7/04, effective 6/1/04)

WAC 388-475-0550 SSI-related medical—All other excluded resources. All resources described in this section are excluded resources for SSI-related medical programs. Unless otherwise stated, interest earned on the resource amount is counted as unearned income.

- (1) Resources necessary for a client who is blind or disabled to fulfill a department approved self-sufficiency plan.
- (2) Retroactive payments from SSI or RSDI, including benefits a client receives under the interim assistance reimbursement agreement with the Social Security Administration, are excluded for ((six)) nine months following the month of receipt. This exclusion applies to:
- (a) Payments received by the client, spouse, or any other person financially responsible for the client;
- (b) SSI payments for benefits due for the month(s) before the month of continuing payment;
- (c) RSDI payments for benefits due for a month that is two or more months before the month of continuing payment; and
- (d) Proceeds from these payments as long as they are held as cash, or in a checking or savings account. The funds may be commingled with other funds, but must remain identifiable from the other funds for this exclusion to apply. This exclusion does not apply once the payments have been converted to any other type of resource.
- (3) All resources specifically excluded by federal law, such as those described in subsections (4) through (11) as long as such funds are identifiable.
- (4) Payments made under Title II of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970
- (5) Payments made to Native Americans as listed in 20 CFR 416.1182, Appendix to subpart K, section IV, paragraphs (b) and (c), and in 20 CFR 416.1236.

Emergency [14]

- (6) The following Native American/Alaska Native funds are excluded resources:
- (a) Resources received from a Native Corporation under the Alaska Native Claims Settlement Act, including:
- (i) Shares of stock held in a regional or village corporation;
- (ii) Cash or dividends on stock received from the Native Corporation up to two thousand dollars per person per year;
- (iii) Stock issued by a native corporation as a dividend or distribution on stock;
  - (iv) A partnership interest;
  - (v) Land or an interest in land; and
  - (vi) An interest in a settlement trust.
- (b) All funds contained in a restricted Individual Indian Money (IIM) account.
- (7) Restitution payment and any interest earned from this payment to persons of Japanese or Aleut ancestry who were relocated and interned during war time under the Civil Liberties Act of 1988 and the Aleutian and Pribilof Islands Restitution Act.
- (8) Funds received from the Agent Orange Settlement Fund or any other funds established to settle Agent Orange liability claims.
- (9) Payments or interest accrued on payments received under the Radiation Exposure Compensation Act received by the injured person, the surviving spouse, children, grandchildren, or grandparents.
  - (10) Payments from:
- (a) The Dutch government under the Netherlands' Act on Benefits for Victims of Persecution (WUV).
- (b) The Victims of Nazi Persecution Act of 1994 to survivors of the Holocaust.
- (c) Susan Walker vs. Bayer Corporation, et al., 96-C-5024 (N.D. Ill.) (May 8, 1997) settlement funds.
- (d) Ricky Rey Hemophilia Relief Fund Act of 1998 P.L. 105-369.
- (11) The unspent social insurance payments received due to wage credits granted under sections 500 through 506 of the Austrian General Social Insurance Act.
- (12) Earned income tax credit refunds and payments are excluded as resources ((during the month of receipt and the following month)) for nine months after the month of receipt.
- (13) Payments from a state administered victim's compensation program for a period of nine calendar months after the month of receipt.
- (14) Cash or in-kind items received as a settlement for the purpose of repairing or replacing a specific excluded resource are excluded:
- (a) For nine months. This includes relocation assistance provided by state or local government.
  - (b) Up to a maximum of thirty months, when:
- (i) The client intends to repair or replace the excluded resource; and
- (ii) Circumstances beyond the control of the settlement recipient prevented the repair or replacement of the excluded resource within the first or second nine months of receipt of the settlement.
- (c) For an indefinite period, if the settlement is from federal relocation assistance.

- (d) Permanently, if the settlement is assistance received under the Disaster Relief and Emergency Assistance Act or other assistance provided under a federal statute because of a catastrophe which is declared to be a major disaster by the President of the United States, or is comparable assistance received from a State or local government or from a disaster assistance organization. Interest earned on this assistance is also excluded from resources. Any cash or in-kind items received as a settlement and excluded under this subsection are considered as available resources when not used within the allowable time periods.
- (15) Insurance proceeds or other assets recovered by a Holocaust survivor as defined in WAC 388-470-0026(4).
- (16) Pension funds owned by an ineligible spouse. Pension funds are defined as funds held in a(n):
- (a) Individual retirement account (IRA) as described by the IRS code; or
- (b) Work-related pension plan (including plans for selfemployed individuals, known as Keogh plans).
- (17) Cash payments received from a medical or social service agency to pay for medical or social services are excluded for one calendar month following the month of receipt.
- (18) SSA- or DVR-approved plans for achieving self-support (PASS) accounts, allowing blind or disabled individuals to set aside resources necessary for the achievement of the plan's goals, are excluded.
- (19) Food and nutrition programs with federal involvement. This includes Washington Basic Food, school reduced and free meals and milk programs and WIC.
- (20) Gifts to, or for the benefit of, a person under eighteen years old who has a life-threatening condition, from an organization described in section 501 (c)(3) of the Internal Revenue Code of 1986 which is exempt from taxation under section 501(a) of that Code, as follows:
  - (a) In-kind gifts that are not converted to cash; or
- (b) Cash gifts up to a total of two thousand dollars in a calendar year.
- (((22))) (21) Veteran's payments made to, or on behalf of, natural children of Vietnam veterans regardless of their age or marital status, for any disability resulting from spina bifida suffered by these children.
- $(((\frac{23}{2})))$  (22) The following are among assets that are not considered resources and as such are neither excluded nor counted:
- (a) Home energy assistance/support and maintenance assistance;
- (b) Retroactive in-home supportive services payments to ineligible spouses and parents; and
- (c) Gifts of domestic travel tickets. For a more complete list please see POMS @ http://policy.ssa.gov/poms.nsf/lnx/0501130050.

AMENDATORY SECTION (Amending WSR 04-09-004, filed 4/7/04, effective 6/1/04)

WAC 388-475-0700 SSI-related medical—Income eligibility. (1) In order to be eligible, a client is required do everything necessary to obtain any income to which they are entitled including (but not limited to):

[15] Emergency

- (a) Annuities,
- (b) Pensions,
- (c) Unemployment compensation,
- (d) Retirement, and
- (e) Disability benefits; even if their receipt makes the client ineligible for department services, unless the client can provide evidence showing good reason for not obtaining the benefits.

The department does not count this income until the client begins to receive it.

- (2) Income is budgeted prospectively for all medical programs.
- (3) Anticipated nonrecurring lump sum payments other than retroactive SSI/SSDI payments are considered income in the month received, subject to reporting requirements in WAC 388-418-0007(4). Any unspent portion is considered a resource the first of the following month.
- (4) The department follows income and resource methodologies of the Supplemental Security Income (SSI) program defined in federal law when determining eligibility for SSI-related medical or Medicare Savings programs unless the department adopts rules that are less restrictive than those of the SSI program.
  - (5) Exceptions to the SSI income methodology:
- (a) Lump sum payments from a retroactive SSDI benefit, when reduced by the amount of SSI received during the period covered by the payment, are not counted as income;
- (b) Unspent retroactive lump sum money from SSI or SSDI is excluded as a resource for ((six)) nine months following receipt of the lump sum; and
- (c) Both the principal and interest portions of payments from a sales contract, that meet the definition in WAC 388-475-0350(10), are unearned income.
- (6) To be eligible for categorically needy (CN) SSI-related medical coverage, a client's countable income cannot exceed the CN program standard described in:
- (a) WAC 388-478-0065 through 388-478-0085 for non-institutional medical unless living in an alternate living facility; or
- (b) WAC 388-513-1305(2) for noninstitutional CN benefits while living in an alternate living facility; or
- (c) WAC 388-513-1315 for institutional and waiver services medical benefits.
- (7) To be eligible for SSI-related medical coverage provided under the medically needy (MN) program, a client must:
- (a) Have countable income at or below the MN program standard as described in WAC 388-478-0070; or
- (b) Satisfy spenddown requirements described in WAC 388-519-0110((<del>\_ or</del>));
- (c) Meet the requirements for noninstitutional MN benefits while living in an alternate living facility (ALF). See WAC 388-513-1305(3) ((and 388-515-1540)); or
- (d) Meet eligibility for the MN waiver program. See WAC 388-515-1540 and 388-515-1550.

AMENDATORY SECTION (Amending WSR 04-09-005, filed 4/7/04, effective 6/1/04)

- WAC 388-475-0800 SSI-related medical—General income exclusions. The department excludes, or does not consider, the following when determining a client's eligibility for SSI-related medical programs:
- (1) The first twenty dollars per month of unearned income. If there is less than twenty dollars of unearned income in a month, the remainder is excluded from earned income in that month.
- (a) The twenty-dollar limit is the same, whether applying it for a couple or for a single person.
- (b) The disregard does not apply to income paid totally or partially by the federal government or a nongovernmental agency on the basis of an eligible person's needs.
- (c) The twenty dollars disregard is applied after all exclusions have been taken from income.
- (2) Income that is not reasonably anticipated or is received infrequently or irregularly, whether for a single person or each person in a couple when it is:
- (a) Earned and does not exceed a total of ((ten)) thirty dollars per ((month)) calendar quarter; or
- (b) Unearned and does not exceed a total of ((twenty)) sixty dollars per ((month)) calendar quarter;
- (c) Increases in a client's burial funds that were established on or after November 1, 1982 if the increases are the result of:
  - (i) Interest earned on excluded burial funds; or
- (ii) Appreciation in the value of an excluded burial arrangement that was left to accumulate and become part of separately identified burial funds.
- (3) Essential expenses necessary for a client to receive compensation (e.g., necessary legal fees in order to get a settlement);
- (4) Receipts, which are not considered income, when they are for:
  - (a) Replacement or repair of an exempt resource;
- (b) Prepayment or repayment of medical care paid by a health insurance policy or medical service program; or
- (c) Payments made under a credit life or credit disability policy.
- (5) The fee a guardian or representative payee charges as reimbursement for providing services, when such services are a requirement for the client to receive payment of the income
  - (6) Funds representing shared household costs.
  - (7) Crime victim's compensation.
- (8) The value of a common transportation ticket, given as a gift, that is used for transportation and not converted to cash
- (9) Gifts that are not for food, clothing or shelter, and gifts of home produce used for personal consumption.
- (10) The department does not consider in-kind income received from someone other than a person legally responsible for the individual unless it is earned. Therefore, the following in-kind payments are not counted when determining eligibility for SSI-related medical programs.
- (a) In-kind payments for services paid by a client's employer if:

Emergency [16]

- (i) The service is not provided in the course of an employer's trade or business; or
  - (ii) It is in the form of food and/or shelter that is:
  - (A) On the employer's business premises;
  - (B) For the employer's convenience; and
- (C) If shelter, acceptance by the employee is a condition of employment.
- (b) In-kind payments made to people in the following categories:
  - (i) Agricultural employees;
  - (ii) Domestic employees;
  - (iii) Members of the Uniformed Services;
- (iv) Persons who work from home to produce specific products for the employer from materials supplied by the employer.

### AMENDATORY SECTION (Amending WSR 04-09-005, filed 4/7/04, effective 6/1/04)

- WAC 388-475-0820 SSI-related medical—Child-related income exclusions. (1) The department excludes an allowance from a person's earned and/or unearned income for a child living in the home when:
  - (a) The minor child lives with an SSI-related parent; and
- (b) The minor child is not receiving a needs-based cash payment such as TANF or SSI; and
  - (c) The SSI-related parent is single; or
- (d) The SSI-related parent lives with a spouse who has no income; and
- (e) The individual applying for or receiving SSI-related medical benefits is the adult parent. The maximum allowance is one-half the Federal Benefit Rate (FBR) for each child. The child's countable income, if any, is subtracted from the maximum child's allowance((-One third of the child support received for the child is excluded from the child's income)) before determining this allowance.
- (2) Foster care payments received for a child who is not SSI-eligible and who is living in the household, placed there by a licensed, nonprofit or public child placement or child-care agency are excluded from income regardless of whether the person requesting or receiving SSI-related medical is the adult foster parent or the child who was placed.
- (3) Adoption support payments, received by an adult for a child in the household that are designated for the child's needs, are excluded as income. Adoption support payments that are not specifically designated for the child's needs are not excluded and are considered unearned income to the adult.
- (4) ((Up to one thousand three hundred seventy dollars per month of a child's)) Earned income((, but not more than five thousand five hundred twenty dollars per year,)) of a person under age twenty-two is excluded if ((the child)) that person is a student.
- (5) Child support payments received from an absent parent for a child living in the home are considered the income of the child.
- (6) One-third of child support payments received for a child are excluded from the child's income.
- (7) Any portion of a grant, scholarship, ((e+)) fellowship, or gift used ((to pay)) for tuition, fees and/or other necessary

- educational expenses at any educational institution is excluded from income for nine months after the month of receipt.
- (((<del>7)</del>)) (<u>8</u>) Gifts to, or for the benefit of, a person under eighteen years old who has a life-threatening condition, from an organization described in section 501 (c)(3) of the Internal Revenue Code of 1986 which is exempt from taxation under section 501(a) of that Code, is excluded as follows:
  - (a) In-kind gifts that are not converted to cash; or
- (b) Cash gifts up to a total of two thousand dollars in a calendar year.
- (((8))) (9) Veteran's payments made to, or on behalf of, natural children of Vietnam veterans regardless of their age or marital status, for any disability resulting from spina bifida suffered by these children are excluded from income.
- ((<del>(9)</del>)) (10) Unless it is specifically contributed to the client, all earned income of an ineligible or nonapplying person under the age of twenty-one who is a student:
  - (a) Attending a school, college, or university; or
- (b) Pursuing a vocational or technical training program designed to prepare the student for gainful employment.

### AMENDATORY SECTION (Amending WSR 04-09-005, filed 4/7/04, effective 6/1/04)

- WAC 388-475-0860 SSI-related medical—Income exclusions under federal statute or other state laws. The Social Security Act and other federal statutes or state laws list income that the department excludes when determining eligibility for SSI-related medical programs. These exclusions include, but are not limited to:
  - (1) Income tax refunds;
- (2) Federal earned income tax credit (EITC) payments for nine months after the month of receipt;
- (3) Compensation provided to volunteers in the Corporation for National and Community Service (CNCS), formerly known as ACTION programs established by the Domestic Volunteer Service Act of 1973. P.L. 93-113;
- (4) Assistance to a person (other than wages or salaries) under the Older Americans Act of 1965, as amended by section 102 (h)(1) of Pub. L. 95-478 (92 Stat. 1515, 42 U.S.C. 3020a);
- (5) Federal, state and local government payments including assistance provided in cash or in-kind under any government program that provides medical or social services;
- (6) Certain cash or in-kind payments a client receives from a governmental or nongovernmental medical or social service agency to pay for medical or social services;
- (7) Value of food provided through a federal or nonprofit food program such as WIC, donated food program, school lunch program;
  - (8) Assistance based on need, including:
- (a) Any federal SSI income or state supplement payment (SSP) based on financial need;
  - (b) Food stamps;
  - (c) GA-U;
  - (d) CEAP;
  - (e) TANF; and
  - (f) Bureau of Indian Affairs (BIA) general assistance.

[17] Emergency

- (9) Housing assistance from a federal program such as HUD if paid under:
- (a) United States Housing Act of 1937 (section 1437 et seq. of 42 U.S.C.);
- (b) National Housing Act (section 1701 et seq. of 12 U.S.C.);
- (c) Section 101 of the Housing and Urban Development Act of 1965 (section 1701s of 12 U.S.C., section 1451 of 42 U.S.C.);
- (d) Title V of the Housing Act of 1949 (section 1471 et seq. of 42 U.S.C.); or
  - (e) Section 202(h) of the Housing Act of 1959;
- (f) Weatherization provided to low-income homeowners by programs that consider income in the eligibility determinations:
  - (10) Energy assistance payments including:
  - (a) Those to prevent fuel cutoffs, and
  - (b) To promote energy efficiency.
- (11) Income from employment and training programs as specified in WAC 388-450-0045.
  - (12) Foster Grandparents program;
- (13) Title IV-E and state foster care maintenance payments if the foster child is not included in the assistance unit;
- (14) The value of any childcare provided or arranged (or any payment for such care or reimbursement for costs incurred for such care) under the Child Care and Development Block Grant Act, as amended by section 8(b) of P.L. 102-586 (106 Stat. 5035).
- (15) Educational assistance as specified in WAC 388-450-0035.
- (16) Up to two thousand dollars per year derived from an individual's interest in Indian trust or restricted land.
- (17) Native American benefits and payments as specified in WAC 388-450-0040 and other Native American payments excluded by federal statute. For a complete list of these payments, see 20 CFR 416, Subpart K, Appendix, IV.
- (18) Payments from Susan Walker v. Bayer Corporation, et al., 96-c-5024 (N.D. Ill) (May 8, 1997) settlement funds;
- (19) Payments from Ricky Ray Hemophilia Relief Fund Act of 1998, P.L. 105-369;
- (20) Disaster assistance paid under Federal Disaster Relief P.L. 100-387 and Emergency Assistance Act, P.L. 93-288 amended by P.L. 100-707 and for farmers P.L. 100-387;
- (21) Payments to certain survivors of the Holocaust as victims of Nazi persecution; payments excluded pursuant to section 1(a) of the Victims of Nazi Persecution Act of 1994, P.L. 103-286 (108 Stat. 1450);
- (22) Payments made under section 500 through 506 of the Austrian General Social Insurance Act;
- (23) Payments made under the Netherlands' Act on Benefits for Victims of Persecution (WUV);
- (24) Restitution payments and interest earned to Japanese Americans or their survivors, and Aleuts interned during World War II, established by P.L. 100-383;
- (25) Payments made from the Agent Orange Settlement Funds or any other funds to settle Agent Orange liability claims established by P.L. 101-201;
- (26) Payments made under section six of the Radiation Exposure Compensation Act established by P.L. 101-426;

(27) Any interest ((earned from payments described in subsections (1) through (26) is counted as unearned income, unless otherwise excluded by law)) or dividend is excluded as income, except for the community spouse of an institutionalized individual.

#### WSR 06-05-040 EMERGENCY RULES DEPARTMENT OF FISH AND WILDLIFE

[Order 06-20—Filed February 9, 2006, 4:32 p.m., effective February 9, 2006]

Effective Date of Rule: Immediately.

Purpose: Amend commercial fishing rules.

Citation of Existing Rules Affected by this Order: Repealing WAC 220-52-07300C.

Statutory Authority for Adoption: RCW 77.12.047.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: The 2005-2006 state sea urchin harvest quota share amounts within specific harvest areas have either been taken, or, there is a high likelihood that continued harvest of product would result in significant wastage of low market quality urchins. There is insufficient time to promulgate permanent rules.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 0, Amended 0, Repealed 1.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 0, Repealed 0.

Date Adopted: February 9, 2006.

J. P. Koenings Director

#### **REPEALER**

The following section of the Washington Administrative Code is repealed:

WAC 220-52-07300C Sea urchins. (06-11)

Emergency [18]

#### WSR 06-05-047 EMERGENCY RULES DEPARTMENT OF FISH AND WILDLIFE

[Order 06-18—Filed February 10, 2006, 4:38 p.m., effective February 10, 2006]

Effective Date of Rule: Immediately.

Purpose: Amend personal use fishing rules.

Citation of Existing Rules Affected by this Order: Amending WAC 232-28-619.

Statutory Authority for Adoption: RCW 77.12.047.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: The upriver spring chinook return in the Columbia River is projected to be 88,400. With a projected run of this size, additional opportunity for sport fisheries in the Columbia River is available. The season is consistent with Washington fish and wildlife commission guidance for 2006-2007. The extended season is expected to allow for harvest of hatchery chinook while minimizing impacts to ESA listed species. Rule is consistent with joint state actions of [the] Washington department of fish and wildlife and Oregon department of fish and wildlife on January 26, 2006. There is insufficient time to promulgate permanent rules.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 1, Amended 0, Repealed 0.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 0, Repealed 0.

Date Adopted: February 10, 2006.

J. P. Koenings Director by Larry Peck

#### **NEW SECTION**

WAC 232-28-61900H Exceptions to statewide rules—Columbia River. Notwithstanding the provisions of WAC 232-28-619, it is unlawful to violate the following provisions, provided that unless otherwise amended all permanent rules remain in effect:

#### 1. Columbia River:

i. From a true north-south line through Buoy 10 to the I-5 Bridge: Salmon, steelhead and shad: Open immediately

until further notice. Daily limit 6 salmon of which no more than 2 may be adult salmon. Release all wild Chinook, wild coho, sockeye and chum. Minimum size 12 inches. Daily limit 2 trout, release wild steelhead and wild cutthroat. Minimum size 12-inches.

ii. From Tower Island power lines in Bonneville Pool upstream to McNary Dam, (except for those waters closed under permanent regulations). Salmon, steelhead and shad: Open March 16 until further notice. Daily limit 6 salmon of which no more than 2 may be adult salmon. Release all wild Chinook, sockeye and chum. Minimum size 12 inches. Daily limit 2 trout, release wild steelhead. Minimum size 12-inches.

iii. For the mainstem Columbia River salmon and steel-head fishery upstream of the Rocky Point/Tongue Point line effective during February 15 through May 15, 2006, salmon and steelhead required to be released may not be totally removed from the water, except anglers fishing from vessels thirty feet or longer shown on their state registration or Coast Guard documentation are exempt from this subsection.

## WSR 06-05-048 EMERGENCY RULES DEPARTMENT OF FISH AND WILDLIFE

[Order 06-21—Filed February 10, 2006, 4:39 p.m., effective February 15, 2006, 6:00 p.m.]

Effective Date of Rule: February 15, 2006, 6:00 p.m.

Purpose: Amend commercial use fishing rules.

Citation of Existing Rules Affected by this Order: Repealing WAC 220-52-04600G; and amending WAC 220-52-046.

Statutory Authority for Adoption: RCW 77.12.047.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: This closure complies with state/treaty management agreements which have established closure periods to protect crab during softshell periods. There is insufficient time to promulgate permanent rules.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 1, Amended 0, Repealed 1.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Mak-

[19] Emergency

ing: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 0, Repealed 0.

Date Adopted: February 10, 2006.

J. P. Koenings Director by Larry Peck

#### **NEW SECTION**

WAC 220-52-04600J Crab fishery—Seasons and areas. Notwithstanding the provisions of WAC 220-52-046, effective, 6:00 p.m. February 15, 2006 until further notice:

- 1) It is unlawful to fish for Dungeness Crab for commercial purposes in Marine Fish-Shellfish Management and Catch Reporting Areas 24A, 24B, 24C, 24D and 26A East.
- 2) Effective immediately until further notice, it will be lawful to fish for Dungeness crab for commercial purposes in the following areas:
- (a) Those waters of Marine Fish-Shellfish Management and Catch Reporting Area 26A within a line that extends due north from the green number 1 buoy at Scatchet Head to Scatchet Head, thence from the green number 1 buoy at Scatchet Head to the green number 1 buoy at Possession Point, thence due north from the green number 1 buoy at Possession Point to Possession Point.

#### REPEALER

The following section of the Washington Administrative Code is repealed effective 6:00 p.m. February 15, 2006:

WAC 220-52-04600G Crab fishery—Seasons and areas. (05-282)

#### WSR 06-05-095 EMERGENCY RULES DEPARTMENT OF FISH AND WILDLIFE

[Order 06-22—Filed February 14, 2006, 2:37 p.m., effective March 1, 2006]

Effective Date of Rule: March 1, 2006. Purpose: Amend personal use rules.

Citation of Existing Rules Affected by this Order: Repealing WAC 220-56-35000X and 220-56-38000J; and amending WAC 220-56-350 and 220-56-380.

Statutory Authority for Adoption: RCW 77.12.240.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: This emergency regulation is needed to open the sport clam and oyster season at Penrose State Park. Projected sport effort shows that the clam and oyster resource on this beach can support a longer season. There is insufficient time to promulgate permanent rules.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal

Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 2, Amended 0, Repealed 2.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 0, Repealed 0.

Date Adopted: February 14, 2006.

J. P. Koenings
Director
by Larry Peck

#### **NEW SECTION**

WAC 220-56-35000Y Clams other than razor clams—Areas and seasons. Notwithstanding the provisions of WAC 220-56-350, effective immediately until further notice, it is unlawful to take, dig for and possess clams, cockles, and mussels taken for personal use from the following public tidelands except during the open periods specified herein:

- (1) Penrose Point State Park: Open March 1 through May 31.
  - (2) Rendsland Creek: Open through May 31.

#### **NEW SECTION**

WAC 220-56-38000K Oysters—Areas and seasons Notwithstanding the provisions of WAC 220-56-380, effective immediately until further notice, it is unlawful to take, dig for and possess oysters taken for personal use from the following public tidelands except during the open periods specified herein:

- (1) Penrose Point State Park: Open March 1 through May 31.
  - (2) Sequim Bay State Park: Open until further notice.
- (3) Cushman Park: Closed through April 30. Open May 1 through May 31.

#### **REPEALER**

The following sections of the Washington Administrative Code are repealed effective March 1, 2006:

WAC 220-56-35000X Clams other than razor clams—Areas and seasons. (05-278)

WAC 220-56-38000J Oysters—Areas and sea-

sons. (06-01)

Emergency [20]

# WSR 06-05-116 EMERGENCY RULES DEPARTMENT OF FISH AND WILDLIFE

[Order 06-25—Filed February 15, 2006, 11:26 a.m., effective March 1, 2006]

Effective Date of Rule: March 1, 2006.

Purpose: Amend personal use fishing rules.

Citation of Existing Rules Affected by this Order: Repealing WAC 232-28-61900J; and amending WAC 232-28-619.

Statutory Authority for Adoption: RCW 77.12.047.

Under RCW 34.05.350 the agency for good cause finds that immediate adoption, amendment, or repeal of a rule is necessary for the preservation of the public health, safety, or general welfare, and that observing the time requirements of notice and opportunity to comment upon adoption of a permanent rule would be contrary to the public interest.

Reasons for this Finding: The 2005/2006 wild winter steelhead runs for the Snohomish and Stillaguamish River systems are forecasted to be below escapement needs. Predicted wild steelhead run sizes for the river systems are anticipated to be only 36% of wild steelhead escapement requirements. Since the predicted wild run sizes are so low, closures are necessary to eliminate hooking mortality and disturbance of spawning fish to maximize escapement. There is insufficient time to promulgate permanent rules.

Number of Sections Adopted in Order to Comply with Federal Statute: New 0, Amended 0, Repealed 0; Federal Rules or Standards: New 0, Amended 0, Repealed 0; or Recently Enacted State Statutes: New 0, Amended 0, Repealed 0.

Number of Sections Adopted at Request of a Nongovernmental Entity: New 0, Amended 0, Repealed 0.

Number of Sections Adopted on the Agency's Own Initiative: New 1, Amended 0, Repealed 1.

Number of Sections Adopted in Order to Clarify, Streamline, or Reform Agency Procedures: New 0, Amended 0, Repealed 0.

Number of Sections Adopted Using Negotiated Rule Making: New 0, Amended 0, Repealed 0; Pilot Rule Making: New 0, Amended 0, Repealed 0; or Other Alternative Rule Making: New 0, Amended 0, Repealed 0.

Date Adopted: February 15, 2006.

J. P. Koenings Director

#### NEW SECTION

WAC 232-28-61900J Exceptions to statewide rules— Skykomish River and North Fork Stillaguamish River. Notwithstanding the provisions of WAC 232-28-619:

(1) Skykomish River (Mainstem)

From mouth to mouth of the Sultan River: Closed to fishing March 1, 2006 through April 30, 2006.

(2) Stillaguamish River (North Fork) From mouth to Swede Heaven Bridge: Closed to fishing March 1, 2006 through May

31, 2006.

#### **REPEALER**

The following section of the Washington Administrative Code is repealed effective June 1, 2006:

WAC 232-28-61900J

Exceptions to statewide rules—Skykomish River and North Fork Stillaguamish River.

[21] Emergency